Author’s response to reviews

Title: Tissue oxygenation as a target for goal-directed therapy in high-risk surgery: a pilot study

Authors:

Paul A van Beest (p.van.beest@umcg.nl)
Jaap Jan Vos (j.j.vos@umcg.nl)
Marieke Poterman (m.poterman@umcg.nl)
Alain F Kalmar (a.kalmar@umcg.nl)
Thomas WL Scheeren (t.w.l.scheeren@umcg.nl)

Version: 4
Date: 1 December 2014

Author’s response to reviews: see over
Groningen, 24.11.2014

BMC Anesthesiology 1214711772137483

Tissue oxygenation as a target for goal-directed therapy in high-risk surgery: a pilot study

Dear editor,

Again, we would like to thank you and the reviewers for the positive response to our manuscript, and for the excellent feedback, which we thought was most constructive and helpful.

As mentioned in your last correspondence the manuscript is now accepted, after one editorial requirement that is.

As requested, the Trial Registration Number is now mentioned at the end of the abstract, including the date of registration.

On behalf of all co authors,

Yours sincerely,

Paul van Beest, MD PhD